Author's response to reviews

Title: Assessment of a Continuous Blood Gas Monitoring System in Animals during Circulatory stress.

Authors:

Sandro Gelsomino (sandro.gelsomino@libero.it)
Roberto Lorusso (roberto_lorusso@iol.it)
Stefano Romagnoli (romagnoliste@tiscali.it)
Ugolino Livi (livi.ugo@aoud.sanita.fvg.it)
Rocco Carella (carella@unich.it)
Fabiana Lucà (fabiana.luca@alice.it)
Francesco Matteucci (melampo6@gmail.com)
Attilio Renzulli (renzulli@unicz.it)
Gil Bolotin (sg_bolotin@rambam.health.gov.il)
Giuseppe De Cicco (giudeci@libero.it)
Pierluigi Stefàno (pierluigi.stefano@alice.it)
Jos Maessen (j.maessen@scpc.azm.nl)
Gian Franco Gensini (gfgensini@gmail.com)

Version: 4 Date: 13 December 2010

Author's response to reviews:

We checked that the abstract in the manuscript matched with that of the submission system.
On the title page was added "All Authors read and approved the final manuscript.
The references 10 and 38 were cited.
Title characters were changed as requested.
In the abstract "Introduction" was replaced with "Background".
In the first section of the manuscript "Introduction" was replaced with "Background".
All authors were named in the References.
"Fig" was replaced with Figure.